Author's response to reviews

Title: Implications of screening and childcare exclusion policies for children with Shiga-Toxin producing Escherichia coli infections: Lessons learned from an outbreak in a daycare centre, Norway, 2012

Authors:

Emily MacDonald (emily.macdonald@fhi.no)
Per Kjetil Dalane (per.kjetil.dalane@vensesla.kommune.no)
Preben Aavitsland (preben@epidemi.no)
Lin Thorstensen Brandal (LinThorstensen.Brandal@fhi.no)
Astrid Louise Wester (Astrid.Louise.Wester@fhi.no)
Line Vold (line.vold@fhi.no)

Version: 2
Date: 18 September 2014

Author's response to reviews:

Please find the comments to the editors and reviewers in the attached document,